Author's response to reviews

Title: Health promotion in primary and secondary schools in Denmark: Time trends and associations with schools' and students' characteristics

Authors:

Kirsten Nabe-Nielsen (nabe@sund.ku.dk)
Rikke Krølner (rkr@si-folkesundhed.dk)
Laust H Mortensen (lamo@sund.ku.dk)
Marie B Jørgensen (mbj@nrcwe.dk)
Finn Diderichsen (fidi@sund.ku.dk)

Version: 4
Date: 15 January 2015

Author's response to reviews:

None of the reviewers had any further comments.

Based on the editor's request, we have made the following changes:

1) We have added the following sentence to the methods section:
"We received permission to use data from the steering committee of the Danish part of the HBSC study."

2) We have edited the authors' contributions according to the instructions:
Authors' contributions
KNN, RK, LHM, MBJ, and FD participated in the formulation of aim and research questions.
KNN and RK performed the statistical analyses.
KNN drafted the manuscript.
KNN, RK, LHM, MBJ, and FD critically reviewed the manuscript and approved the final manuscript.